Author's response to reviews

Title: Apoptosis and fibrosis in the peritoneum induces by gastric cancer cells resulting environment becomes favorable for peritoneal metastases

Authors:

Di Na (dawangmo@yahoo.cn)
Zhi-Dong Lv (houjibofa20080808@163.com)
Fu-Nan Liu (lfn540@126.com)
Yan Xu (xyab@sina.com)
Cheng-Gang Jiang (deanjiangcg@163.com)
Zhe Sun (adam_sun@qq.com)
Zhi-Feng Miao (fengfeng19998@hotmail.com)
Feng Li (dawangmo@yahoo.cn)
Hui-Mian Xu (xuhuimian@yahoo.cn)

Version: 2 Date: 18 July 2011

Author's response to reviews:

We found the editor's comments most helpful and have revised the manuscript accordingly. Thank you for your letter and for the comments concerning our manuscript entitled "Apoptosis and fibrosis in the peritoneum induces by gastric cancer cells resulting environment becomes favorable for peritoneal metastases". We have studied the comments carefully and have made corrections which we hope meet with your approval. As suggested by , a statement of ethics has been included. Revised portions are coloured in red.

I greatly appreciate your help to this paper. I hope that the revised manuscript is now suitable for peer review.